Author's response to reviews

Title: The OPTIMIST-A trial: evaluation of minimally-invasive surfactant therapy in preterm infants 25-28 weeks gestation

Authors:

Peter A Dargaville (peter.dargaville@dhhs.tas.gov.au)
C. Omar F Kamlin (omar.kamlin@thewomens.org.au)
Antonio G De Paoli (antonio.depaoli@dhhs.tas.gov.au)
John B Carlin (john.carlin@mcri.edu.au)
Francesca Orsini (francesca.orsini@mcri.edu.au)
Roger F Soll (roger.soll@vtmednet.org)
Peter G Davis (pgd@unimelb.edu.au)

Version: 4 Date: 5 August 2014

Author's response to reviews:

Dear Editor,

The additional files containing the Ethics approvals and funding confirmation have now been removed as requested.

Yours sincerely

Peter Dargaville

On behalf of all authors